Author's response to reviews

Title: Program level implementation of malaria rapid diagnostic tests (RDTs) use: Outcomes and cost of training health workers at lower level health care facilities in Uganda

Authors:

Daniel J Kyabayinze (drdikyabayinze@yahoo.com)
Carol Asiimwe (carol.assimwe@finddiagnostic.org)
Damalie Nakanjako (drdamalie@yahoo.com)
Jane Nabakoza (jinkscuganda@yahoo.com)
Moses Bajabayite (cmbaja@yahoo.com)
Clare Strachan (c.strachan@malariaconsortium.org)
James K Tibenderana (j.tibenderana@malariaconsortium.org)
Jean-Pierre Van Geetruydan (Jean-Pierre.VanGeertruyden@au.ac.be)

Version: 7 Date: 3 April 2012

Author's response to reviews:

We have made the following changes after the reviewers comments

Reviewer 1: We have added the correct affiliation Number 6 to the last author. We have also added 98% in place of word Majority on the last paragraph of results

Reviewer 2: We have corrected all percentages to no decimal points for digits and put capital H on the name Howthorne

Reviewer 3: No changes were suggested